Author's response to reviews

Title: A survey of attitudes and factors associated with successful CPR knowledge transfer in an older population most likely to witness cardiac arrest: design and methodology

Authors:

Christian Vaillancourt (cvaillancourt@ohri.ca)
Jeremy Grimshaw (jgrimshaw@ohri.ca)
Jamie C Brehaut (jbrehaut@ohri.ca)
Martin Osmond (Osmond@cheo.on.ca)
Manya L Charette (mancharette@ohri.ca)
George A Wells (gawells@ottawaheart.ca)
Ian G Stiell (istiell@ohri.ca)

Version: 2 Date: 2 October 2008

Author's response to reviews:

To whom it may concern:
I am submitting a slightly revised version of this manuscript. Two small changes have been made:

-on page 10 in the last full paragraph, the word "websites" appeared twice. One instance was removed.

-in the Acknowledgements, the statement regarding study funding was revised to reflect joint funding from the Heart and Stroke Foundation of Canada and the Canadian Institutes of Health Research.

I trust these small revisions will not hold up review of this manuscript.

Sincerely,

Christian Vaillancourt